Author's response to reviews

Title: Surgeons agree more on treatment recommendations than on classification of proximal humeral fractures

Authors:

Stig Brorson (sbrorson@hotmail.com)
Bo S Olsen (bosaol01@heh.regionh.dk)
Lars H Frich (frichs@daldnet.dk)
Steen L Jensen (steen.lund.jensen@rn.dk)
Anne Kathrine B Sørensen (ankaso01@heh.regionh.dk)
Michael R Krogsgaard (mrk@daldnet.dk)
Asbjørn Hróbjartsson (ah@cochrane.dk)

Version: 5 Date: 7 June 2012

Author's response to reviews:

Dear Editor,

Our study did not involve human individuals and ethical approval was waived by The Regional Committees on Health Research Ethics of the Capital Region of Denmark (Reference no. H-3-2012-FSP35#).

Pleased find attached (as additional material files) a scanned copy of the letter from the ethics committee confirming that ethical approval was not needed for our study.

This has also been stated in the manuscript.

Thank you again for considering our manuscript for publication in BMC Musculoskeletal Disorders.

Sincerely,

Stig Brorson, MD PhD